Craig Chambers CSE 505 (Fall ’98) Reading Assignments

Reading Assignment #5

“A Proposal for Making Eiffel Type-Safe”:

read about problems with the “nice” covariant redefinition rule, and some fixes

“Pizza into Java”:

An extension of Java with F-bounded parametric polymorphism, first-class functions, and a kind of pattern matching (called “algebraic datatypes” in the paper)

“The Cecil Language: Specification and Rationale”:

concentrate on typechecking of multimethods in section 3 and bounded parametric polymorphism in section 4 (note that section 4 is out of date w.r.t. the current Cecil system; the current type system is more flexible (e.g. signature bounds) than that described in section 4)